

# 

**Citation:** Shaman J, Kandula S, Yang W, Karspeck A (2017) The use of ambient humidity conditions to improve influenza forecast. PLoS Comput Biol 13(11): e1005844. https://doi.org/10.1371/journal.pcbi.1005844

Editor: Mercedes Pascual, University of Chicago, UNITED STATES

Received: May 29, 2017

Accepted: October 24, 2017

Published: November 16, 2017

**Copyright:** © 2017 Shaman et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data used in this study are publicly available from the US Centers for Disease Control and Prevention (https://www.cdc. gov/flu/weekly/) and the National Aeronautics and Space Administration (https://ldas.gsfc.nasa.gov/ nldas/); the codes that support the findings of this study are available at cpid.iri.columbia.edu.

**Funding:** This work was supported by US NIH grants GM100467 and GM110748, as well as NIEHS Center grant ES009089. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the **RESEARCH ARTICLE** 

# The use of ambient humidity conditions to improve influenza forecast

#### Jeffrey Shaman<sup>1</sup>\*, Sasikiran Kandula<sup>1</sup>, Wan Yang<sup>1</sup>, Alicia Karspeck<sup>2</sup>

1 Department of Environmental Health Sciences, Mailman School of Public Health, Columbia University, New York, NY, United States of America, 2 National Center for Atmospheric Research, Boulder, CO, United States of America

\* jls106@cumc.columbia.edu

## Abstract

Laboratory and epidemiological evidence indicate that ambient humidity modulates the survival and transmission of influenza. Here we explore whether the inclusion of humidity forcing in mathematical models describing influenza transmission improves the accuracy of forecasts generated with those models. We generate retrospective forecasts for 95 cities over 10 seasons in the United States and assess both forecast accuracy and error. Overall, we find that humidity forcing improves forecasts performance (at 1–4 lead weeks, 3.8% more peak week and 4.4% more peak intensity forecasts are accurate than with no forcing) and that forecasts generated using daily climatological humidity forcing generally outperform forecasts that utilize daily observed humidity forcing (4.4% and 2.6% respectively). These findings hold for predictions of outbreak peak intensity, peak timing, and incidence over 2-and 4-week horizons. The results indicate that use of climatological humidity forcing is warranted for current operational influenza forecast.

## Author summary

Laboratory and epidemiological evidence indicate that atmospheric absolute humidity conditions modulate the survival, transmission, incidence and seasonality of influenza. Absolute humidity (AH) conditions are often incorporated as a forcing factor in mathematical models used to describe and forecast influenza incidence. Here we examine whether the inclusion of absolute humidity forcing improves influenza forecast accuracy. We perform retrospective influenza forecasting over 10 seasons for 95 cities using 4 different forms of AH forcing: 1) no AH forcing; 2) optimization and forecast with local climatological AH forcing; 3) optimization and forecast with local observed AH forcing; must be find that humidity forcing improves forecast performance and that forecasts generated using climatological humidity forcing generally outperform forecasts that utilize observed humidity forcing.

manuscript. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institute Of General Medical Sciences or the National Institutes of Health.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: JS and AK declare partial ownership of SK Analytics.

### Introduction

A growing body of evidence indicates that the survival and transmissibility of influenza are affected by ambient humidity conditions. Laboratory experiments have shown that aerosolized influenza survival rates increase in low ambient relative humidity (RH), i.e. less than 40% [1–10]. Additional experiments examining the transmission of human influenza among guinea pigs have also shown that transmission increases at low RH levels [11]. Further evaluation of these experiments has revealed a strong relationship in which low *absolute* humidity (AH) conditions favor the survival and transmission of influenza [12].

Low AH conditions manifest outdoors during winter in temperate regions. Furthermore, because temperature indoors is managed but humidity is generally not, both AH and RH tend to be low indoors during winter. Indeed, even though RH is often maximal outdoors during winter, indoor RH often reaches the very low levels (10–40%) favorable for influenza survival and transmission, and indoor AH closely mirrors outdoor levels [13–16].

People in the developed world, such as the US, spend approximately 90% of their time indoors [17]; as a result, influenza transmission is suspected to occur indoors in the developed world. In addition, because both indoor RH and indoor AH are highly co-variable with outdoor AH, either variable can be estimated using *outdoor* AH. Consequently, regardless of whether one considers RH or AH to be the true modulator of influenza survival and transmissibility (and the mechanisms for this modulation remain undetermined), outdoor AH can be used to estimate the effect of humidity on influenza transmission.

A number of epidemiological studies have found associations between outdoor AH and estimates of influenza incidence or influenza-related mortality. Statistical analysis has shown that the onset of influenza outbreaks is associated with anomalously low AH conditions [18]. Low AH levels in temperate regions have also been associated with increased influenza-associated mortality levels [19], increased influenza transmission intensity [20], and increased influenza incidence [21].

Modeling studies have shown that the seasonality of influenza in temperate regions can be reproduced when influenza transmission potential is modulated by observed AH conditions [18]. Furthermore, AH, along with other dynamical processes, can be used to explain the timing of both seasonal and pandemic influenza outbreaks [22-25]. Indeed, even the development of pandemic influenza outbreaks out of season (i.e. during summer) can be understood in the context of ambient seasonal humidity conditions, contact patterns, and population susceptibility. Specifically, ambient humidity sets an upper bound on the transmission potential (a maximal basic reproductive number), contact patterns also influence transmission (e.g. through preferential mixing among certain sub-populations), and population susceptibility reduces transmissibility as it drops. In effect, AH constrains the extent to which an influenza virus strain is capable of sustained transmission during summer and sustained transmission is possible only if population susceptibility to that strain remains sufficiently high. For circulating seasonal influenza strains, higher AH and lower susceptibility conspire to limit sustained transmission of circulating influenza strains during summer; however, the introduction of a new pandemic strain, for which population susceptibility is much higher, can enable sustained transmission during summer [22, 25].

Recently, we developed a number of model-inference systems for the ensemble forecast of seasonal influenza [26–29]. These systems use a compartmental model, such as a susceptible-infected-recovered-susceptible (SIRS) model, that is first optimized prior to forecast using observational estimates of US state and municipal influenza incidence. When first developed, we used an AH-forced SIRS model for these predictions, as this model had been used to describe the seasonality of influenza at state geographic scales in the US [18]. However, when

applied to influenza forecast over larger areas, e.g. US CDC Health and Human Services multistate regions, or subtropical regions, we have heretofore adopted a SIRS form without AH forcing [27, 30]. It remains an open question whether the prediction of influenza in temperate regions is improved by the inclusion of AH forcing. In this paper, we perform retrospective forecasting over 10 seasons for 95 cities using 4 different forms of AH forcing: 1) no AH forcing; 2) optimization and forecast with local climatological AH forcing (i.e. historical average AH conditions on a given day); 3) optimization and forecast with local observed AH forcing (i.e. AH as observed on a given day); and 4) optimization with observed AH forcing and forecast with climatological AH forcing. This effort applies these 4 AH forcing approaches to an established model-inference prediction system that has been used for 5 years to forecast influenza operationally in real-time [27, 31]. We explore whether clear differences in forecast accuracy emerge among these 4 approaches and quantify whether inclusion of AH forcing improves forecast accuracy. We hypothesize, given evidence suggesting ambient AH modifies the survival and transmissibility of influenza, that AH forcing will improve dynamic model influenza forecast.

#### Methods

### Modeling approaches

Forecast of influenza is here generated using compartmental models describing the propagation of influenza through a population, observational estimates of influenza incidence, and data assimilation methods for model optimization [26–29]. Four different compartmental models were used to generate the forecasts. All four forms are perfectly-mixed, absolute humidity-driven compartmental constructs with the following designations: 1) susceptibleinfectious-recovered (SIR); 2) SIRS; 3) susceptible-exposed-infectious-recovered (SEIR); and 4) susceptible-exposed-infectious-recovered-susceptible (SEIRS). The differences among the model forms align with whether waning immunity, which allows recovered individuals to return to the susceptible class, or an explicit period of latent infection (the exposed period) is represented.

As the SEIRS model is the most detailed, we present it here. All other forms are derived by reduction of these equations, which are as follows:

$$\frac{dS}{dt} = \frac{N - S - E - I}{L} - \frac{\beta(t)IS}{N} - \alpha \tag{1}$$

$$\frac{dE}{dt} = \frac{\beta(t)IS}{N} - \frac{E}{Z} + \alpha \tag{2}$$

$$\frac{dI}{dt} = \frac{E}{Z} - \frac{I}{D} \tag{3}$$

where *S* is the number of susceptible people in the population, *t* is time in years, *N* is the population size, *E* is the number of exposed people, *I* is the number of infectious people, *N*-*S*-*E*-*I* is the number of recovered individuals,  $\beta(t)$  is the contact rate at time *t*, *L* is the average duration of immunity, *Z* is the mean latent period, *D* is the mean infectious period, and  $\alpha$  is the rate of travel-related import of influenza virus into the model domain.

The contact rate,  $\beta(t)$ , is given by  $\beta(t) = R_0(t)/D$ , where  $R_0(t)$ , the basic reproductive number, is the number of secondary infections the average infectious person would produce in a fully susceptible population at time *t*. Specific humidity, a measure of absolute humidity (AH), modulates transmission rates within this model by altering  $R_0(t)$  through an exponential

relationship derived from laboratory experiments [12]:

$$R_0(t) = R_{0\min} + (R_{0\max} - R_{0\min})e^{-aq(t)}$$
(4)

where  $R_{0\min}$  is the minimum daily basic reproductive number,  $R_{0\max}$  is the maximum daily basic reproductive number, a = 180 (unitless), and q(t) is the time-varying specific humidity (in kg/kg). The value of a is estimated from the laboratory regression of influenza virus survival upon AH [18]. Simulations were performed with fixed travel-related seeding of I of 0.1 infections per day (1 infection every 10 days).

For each of the 4 compartmental model forms (SIR, SIRS, SEIR, SEIRS), four different approaches were used to test how the incorporation of AH conditions in the model framework affects the accuracy of influenza forecast. Each approach applied AH conditions in a different fashion. In the first approach, 'observed AH', observed local daily AH conditions are applied using Eq 4. This use of AH is not realistic for real-time forecast, as future AH conditions are not known; however, as applied here, it can be used to forecast influenza outbreak characteristics retrospectively and to determine if such information would, in theory, improve forecast accuracy.

For the second approach, 'climatological AH', we use local daily climatological AH conditions in Eq 4. Here, the climatology is based on 24 years (1979–2002) and represents the historical average conditions on a given day for the location at which the model is applied. Due to averaging over many years, climatological AH is much smoother than observed AH conditions and has been used operationally to generate real-time influenza forecasts [27, 31].

The third approach, 'combination AH', is a hybrid of the first two approaches. Observed AH is used during model optimization prior to forecast, but the forecast of future outcomes is generated using climatological AH. This strategy can be used for real-time forecasting and was implemented previously [26].

The final approach, 'no AH', replaces Eq 4 with  $R_0(t) = R_0$ . In this fashion  $R_0$  is treated as an adjustable parameter to be optimized during the data assimilation process (see below). As with the other parameters of the model, it remains fixed during forecast when the optimized model is integrated into the future to generate an ensemble of predictions.

#### Data

Specific humidity (*q*; used in Eq 4) data were compiled from the National Land Data Assimilation System (NLDAS) project-2 dataset. These data are derived through spatial interpolation, temporal disaggregation and vertical adjustment from station measurements and National Center for Environmental Prediction North American Regional Reanalysis [32]. The gridded NLDAS meteorological data are available in hourly time steps on a 0.125° regular grid from 1979 through the present [33]. Specific humidity data from the grid cell containing the centroid of each of the 95 cities included in this study were assembled for 1979–2015. These hourly data were then averaged to daily resolution. A 1979–2002 (24 year) daily climatology was then constructed for each city.

As described in Shaman et al. [27], weekly estimates of influenza incidence were generated by multiplying 2003–2015 historical Google Flu Trend (GFT) estimates of municipal influenza-like illness (ILI) [34], as these data were released in real time (S1 Data), by coincident census division (regional) weekly laboratory-confirmed influenza positive proportions as compiled through the National Respiratory and Enteric Virus Surveillance System (NREVSS) and U.S.-based World Health Organization (WHO) Collaborating Laboratories [35]. This combined metric, termed ILI+, provides a more specific measure of influenza incidence than ILI alone, which non-specifically captures signal from other circulating respiratory viruses, such as adenovirus and rhinovirus [27, 36]. While the spatially coarser regional NRVESS/WHO data does not fully discriminate variability in influenza positivity at the municipal scale, multiplication of municipal ILI with these regional proportions does remove some of the signal associated with other respiratory viruses and provides a more precise estimate of influenza activity than ILI alone.

#### Data assimilation

Three ensemble filter methods—the ensemble Kalman filter [37], the ensemble adjustment Kalman filter [38] and the rank histogram filter [39]—and a particle filter (PF) with resampling and regularization [40] were used in conjunction with ILI+ to optimize and initialize the compartmental models prior to forecast. Both the model state variables (*S*, *E*, *I*) and parameters (*L*, *Z*, *D*,  $R_{0max}$ ,  $R_{0min}$  and  $R_0$ ) were subject to optimization. Ensemble filter simulations were run with 300 ensemble members and PF simulations were run with 10,000 particles.

Each ensemble filter algorithm is applied sequentially through time to update ensemble model simulations of observed state variables (i.e. influenza incidence) to better align with observations (i.e. ILI+). These updates are calculated, per the specifics of each filter algorithm (described below), by halting the ensemble integration when a new observation comes available. The posterior is then integrated through time using the model equations to the next observation and the process is repeated. Through this iterative updating process the ensemble of simulations provides an increasingly accurate estimate of the observed state variable (i.e. influenza incidence), and estimates of the unobserved variables and parameters (e.g. susceptibility and mean infectious period) are obtained through additional adjustments that take advantage of their co-variability with the observed state variable.

In general, Kalman filters assume normality of the observational error, the prior distribution, and the posterior distribution. Differences among the ensemble filter algorithms manifest from the means by which the update is specified. The ensemble Kalman filter (EnKF) is a stochastic, perturbed observation Kalman filter in which the update of each ensemble member is computed using the current observation plus Gaussian random noise [41]. That is, the posterior for each ensemble member is simply the weighted sum of the prior for that ensemble member and the observation plus random noise with variance equal to the observational error variance. The weights themselves are calculated as ratios of the ensemble prior variance and the observational error variance.

The ensemble adjustment Kalman filter (EAKF) employs a deterministic algorithm to compute the ensemble posterior mean and variance [38]. At each update, the EAKF algorithm aligns the first two ensemble posterior moments with those predicted by Bayes' theorem.

Unlike the EnKF and EAKF, the rank histogram filter (RHF) does not impose a Gaussian structure on the prior, observations or posterior [39]; rather, this filter employs an algorithm that creates an approximate probability distribution by ordering (i.e. ranking) the ensemble prior. In this fashion, the RHF admits non-Gaussian distributions, thus relaxing the normality assumption inherent to most Kalman filters.

For the ensemble filters, multiplicative inflation [38] was applied following the assimilation of each weekly observation of ILI+. The inflation was used to counter the ensemble filter tendency toward 'filter divergence', which occurs when the prior ensemble spread becomes spuriously low. In the absence of inflation, the system may give too little weight to the observations and thus diverge from the true trajectory.

Unlike the above ensemble filters, PFs are an alternate class of assimilation method that do not require assumptions about linearity or normality. The PF approach used here adopts sequential importance sampling with resampling and regularization [40, 42]. Resampling

generates a new suite of particles with equal weight during the model integration whenever the effective sample size is low. Regularization jiggles the state and parameter values of each resampled particle to eliminate redundancies and further sample parameter space around each previously highly weighted particle. As a consequence of resampling and regularization, a much richer range of parameter and state space is spanned than with a basic PF, which relies only on the initial parameter choices.

For all 16 model-filter combinations (i.e. 4 models  $\times$  4 filters) a scaling factor was employed to convert ILI+ to influenza incidence, the quantity represented in the compartmental models, per Shaman et al. [27]. Both ILI and ILI+ are biased in that they only capture persons seeking medical attention and are measured per 100,000 patient visits. The scaling factor partially compensates for this bias by accounting for the probability that a person with influenza seeks medical attention and the probability that a person seeks medical attention for any reason [27]. Even after this scaling, bias in the data likely remains; however, as long as this bias remains stationary, it should not corrupt forecasting. That is, if a given model is well optimized using biased observations, it should make biased predictions, as the accuracy of those out-of-sample predictions is being assessed using the same biased dataset.

Additional details on the application of the ensemble filters and PF to infectious disease models are provided in Shaman and Karspeck [26] and Yang et al. [28]. <u>S1–S3</u> Figs present example forecasts from the more than 1.2 million predictions generated.

#### Forecast metrics

For each model-filter combination, we generated weekly retrospective forecasts of influenza outbreak characteristics during weeks 6–25 of the influenza season (beginning early October) over 10 seasons (2003–2004 through 2014–2015, excluding the pandemic seasons 2008–2009 and 2009–2010). For the ensemble filters, the ensemble mean trajectory was used for forecast accuracy assessment; for the PF, the particle weighted average trajectory was used. A simple average of these trajectory forecasts as generated by all 16 model-filter combinations was used for accuracy analysis unless otherwise specified. We limit our analysis to forecasts made 0–8 weeks before the predicted peak week.

Let O(t) be the ILI+ observed at time t and  $F_w(t)$  the ILI+ forecast made for time t using ILI+ available through week w, i.e. w < t. The predicted peak intensity at w is defined as the maximum of the average forecast trajectory, and the peak week is the week when that maximum occurs. A predicted peak week is defined as *accurate* if it is within ±1 week of the observed peak week, and the predicted peak intensity is deemed accurate if it is within ± 25% of the observed peak intensity. In addition, absolute error was calculated for each prediction of peak week and peak intensity and used to rank the weekly performances of the 4 AH forecast approaches.

Root mean squared error (RMSE) was calculated over the entire trajectory at time horizons of 2 and 4 weeks:

$$RMSE_{w}^{h} = \frac{\sqrt{\sum_{t=w+1}^{w+h} (F_{w}(t) - O(t))^{2}}}{h}$$
(5)

where  $h \in \{2, 4\}$  weeks. We also performed a Friedman test followed by a Nemenyi test to assess whether forecast error differed significantly among the 4 AH forcing approaches. The Friedman test is a non-parametric test that ranks the error of each group—here, each AH forcing approach—for each forecast location-week. The Nemenyi test assesses for statistically significant differences between each pair of ranked groupings.

### Results

For predictions of peak intensity, forecasts using one of the 3 AH forcing approaches were superior at all lead times (**Fig 1**). Peak week forecasts showed similar results, with the exception of the forecasts with 6- and 7-week leads. Note that the number of no AH forecasts generated with 6- and 7-week leads was small, so this result may be due to sampling error. The climatological AH forecasts were most accurate for leads of 0 through 3 weeks. These findings were insensitive to the choice of error margin used to define accuracy (S4 and S5 Figs).

Forecast error rank for peak intensity reveals the climatological AH forecasts most often had the lowest rank (smallest error) at most leads and the no AH forecasts most often had the largest error at all leads (Fig 2). For predictions of peak week, the no AH forecasts again most often had the largest error at all leads; however, differences among the 3 humidity-forced forecasts were less clearly discernible. These findings were confirmed by Friedman rank, which showed that overall the no AH forecasts ranked the worst among the four AH approaches (Table 1). Results from the pairwise comparison revealed highly significant differences among most pairings (p < 0.001) with the exception of the climatological AH-combination AH pair (Table 2).





Humidity forcing - Climatological AH - Combination AH - Observed AH - No AH

Fig 1. Percentage of forecasts accurate for predictions of peak intensity (top, within ±25% of observed peak intensity) and peak timing (bottom, within ±1 week of the observed peak) plotted as a function of forecast lead relative to the predicted peak. Shown are the forecast accuracies for models with climatological AH forcing (green), observed AH forcing (red), a combination of observed AH during optimization and climatological AH during forecast (blue), and no AH forcing (grey). The number of forecasts (log transformed) at each lead is represented by the size of the dot.

https://doi.org/10.1371/journal.pcbi.1005844.g001



Fig 2. Heat map of forecast error rank for predictions of peak intensity (top) and peak timing (bottom) plotted as a function of forecast lead relative to the predicted peak. Weekly forecasts for a location were ranked (1–4) based on prediction error for a given metric, where 1 was the forecast with the least error. Color indicates the number of forecasts at each lead with a given error ranking relative to the other forms. Darker colors indicate more forecasts at a given lead with a particular ranking.

https://doi.org/10.1371/journal.pcbi.1005844.g002

Similar findings hold for RMSE over 2- and 4-week prediction horizons (**Fig 3**). Here, the no AH forecasts rank either best or worst depending on lead time. Specifically, the no AH forecasts rank worst at longer leads but often rank best at shorter leads (2 to 4 weeks). In contrast, the combination and observed AH forecasts most often cluster at ranks 2 and 3. The climato-logical AH forecasts appear to have the greatest diversity of ranking; however, overall it has the lowest mean ranking (**Table 1**) and provides a significant benefit over the 3 other approaches (**Table 2**).

Table 1. Mean Friedman ranks of forecast error for predictions of peak intensity, peak week and incidence during the first 2 weeks (RMSE2) and 4 weeks (RMSE4) of forecast. For pairwise tests of significance see Table 2. Best performing model forms are in bold. Note, two forms may be best if not statistically different.

Forecast	Peak Intensity	Peak Week	RMSE2	RMSE4
Climatological AH	2.36	2.33	2.35	2.36
Combination	2.32	2.34	2.42	2.43
No AH	2.89	2.88	2.69	2.64
Observed AH	2.43	2.45	2.54	2.56

https://doi.org/10.1371/journal.pcbi.1005844.t001

Table 2. Pairwise p-values derived from Nemenyi tests of the forecast ranks shown in Table 1. Asterisks designate differences significant at p<0.01 (\*\*) and p<0.001 (\*\*\*).

Peak Intensity Forecasts			Peak Week Forecasts				
	Climatological	Combination	No AH		Climatological	Combination	No AH
Combination	0.234	-	-	Combination	0.899	-	-
No AH	<0.001***	<0.001***	-	No AH	<0.001***	<0.001***	-
Observed	<0.01**	<0.001***	<0.001***	Observed	<0.001***	<0.001***	<0.001***
RMSE during the first 2 weeks of Forecast (RMSE2)			RMSE during th	RMSE during the first 4 weeks of Forecast (RMSE4)			
	Climatological	Combination	No AH		Climatological	Combination	No AH
Combination	0.003**	-	-	Combination	0.003**	-	-
No AH	<0.001***	<0.001***	-	No AH	<0.001***	<0.001***	-
Observed	<0.001***	<0.001***	<0.001***	Observed	<0.001***	<0.001***	0.001***

https://doi.org/10.1371/journal.pcbi.1005844.t002

When we stratify the forecasts by model form, we find similar results for peak intensity (**Fig 4**). Specifically, the no AH forecasts are less accurate than the 3 AH-forced forecasts for each of the 4 compartmental model forms. Similarly, the forecasts of peak week stratified by model form are similar to the overall findings (**Fig 1**).

For peak intensity, stratification by filter reveals a less clear distinction between the no AH forecasts and the 3 AH-forced forecasts (Fig 5). The EAKF and EnKF no AH peak intensity



Fig 3. As for Fig 2, but showing RMSE of incidence for the first 2 weeks of forecast (top, RMSE 2) and the first 4 weeks (bottom, RMSE 4).

https://doi.org/10.1371/journal.pcbi.1005844.g003





https://doi.org/10.1371/journal.pcbi.1005844.g004

forecasts do comparatively well for 0- to 2-week leads, whereas the RHF no AH forecasts do well for 4- to 8-week leads. The PF no AH forecasts of peak intensity generally perform less well at all leads but 0 weeks. For peak week, the results by filter are less distinct. Here the EAKF no AH forecasts perform better at longer lead times (5–8 weeks), and the EnKF at 0 to 1 week leads.

#### Discussion

Overall, our findings indicate that peak intensity forecast accuracy improves with some type of AH forcing regardless of lead time. With the exception of long-lead time forecasts, where the low number of samples for the no AH approach make it difficult to interpret the results, forecast of peak week timing is also generally improved with AH forcing. Analyses of forecast error reveal that predictions generated with no AH forcing are most likely to perform worse than counterpart AH-forced prediction. While this tendency is consistent for forecast peak intensity, for peak timing, RMSE2 and RMSE4, the no AH approach at times produced superior forecasts (Figs 2 and 3), although the overall performance of the no AH form was worse (Table 1).





#### https://doi.org/10.1371/journal.pcbi.1005844.g005

Among the 3 AH forcing approaches the differences were less clear. The climatological AH approach most often performed best (Table 1); however, for forecast of peak intensity and peak timing, the accuracy of the climatological and combination AH approaches was not statistically different (Table 2). It is interesting that forecasts with climatological AH forcing were more accurate than those generated using observed AH conditions. This finding indicates that the short-term fluctuations of observed AH due to synoptic variability—the 1–4 day variability associated with storms and frontal systems—may actually degrade prediction. Whether this effect is due to the transience of the synoptic AH signal, which corrupts filter optimization, ILI + observational noise, or the simplicity of the models, which may not appropriately represent the effects of these fluctuations on virus transmissibility, is not clear.

To test whether model simplicity underlies our findings, we performed a synthetic test in which we used local daily, observed AH data for 2003–2015 and the SIRS model to generate time series of influenza incidence for 61 cities (see <u>S1 Text</u>). Daily cases were aggregated by calendar week and observations were drawn from a negative binomial distribution. These synthetic observations were then used, in turn, in conjunction with the EAKF, EnKF and RHF filters to optimize the SIRS model, with each of the 4 AH approaches, and generate forecasts of

the synthetic truth for each of the 61 cities and 10 seasons. Forecast results for this synthetic target were similar to the findings for the ILI+ target (<u>S1</u> and <u>S2</u> Tables). In particular, forecasts made using the climatological AH approach most consistently ranked first or tied for first even though observed AH had been used to generate the synthetic observations. This finding indicates that model simplicity is not the primary reason for our findings.

For shorter lead predictions (1 to 4 weeks) the improvement of the climatological AH approach over other approaches appears to be greater for the 2 models that include a latent period (i.e. the SEIR and SEIRS models, see Fig 4). Laboratory evidence and physical considerations point to an immediate effect of ambient humidity on virus survival and transmissibility [4]. However, given that there exists an intrinsic delay between infection and seeking medical attention, a lag between humidity anomalies and observed changes in rates of ILI and influenza positivity rates should be evident, and has similarly been observed for excess pneumonia and influenza mortality [18]. For ILI+, observations are summed over a calendar week, so some individuals are no doubt infected and seek clinical care in the same week; however, for other individuals, the effects of humidity at the end of one week, would be expected to modulate ILI+ levels during the following week [43]. It thus may be that the models with a latent period (i.e. SEIR and SEIRS) are able to capture some of this lag—but perhaps for the wrong reason in that the period of latency accounts for some of the time between infection and seeking medical attention.

While there is no perfect estimate of influenza infection rates, the use of 2003–2015 historical GFT municipal ILI estimates in this study may have introduced some biases. Firstly, Google repeatedly altered their algorithm for estimating ILI [44–45]; we here used GFT ILI estimates as they were initially made available in real time, rather than subsequently revised estimates, in order to produce retrospective forecasts as they would have been generated in real time. Secondly, GFT ILI national and regional estimates have documented inaccuracies relative to target CDC ILI measurements [46–47]. Thirdly, GFT ILI estimates were developed and validated using national and regional CDC ILI targets. Detailing of the precise algorithm used by Google or the process of extrapolation to sub-regional spatial scales has not been published. Our use of municipal ILI estimates, which are spatially correlated and derived from the same GFT algorithm, suggests that the 95 cities used in this study may not provide 95 independent tests of the importance of humidity. Consequently, it is possible that the significance of the rank differences reported in Table 2 are inflated.

In general, our findings support the continued use of climatological AH forcing when generating influenza forecasts. This approach has been our primary means for generating operational real-time forecasts. Many other groups are also presently developing and generating influenza forecasts using a combination of dynamical and/or statistical methodologies [48– 52], both of which have their strengths. Given the findings here, which indicate that inclusion of humidity forcing improves forecast accuracy, this forcing should be included and tested in conjunction with other forecasting systems. Further, as new, improved model forms and filtering methods are brought online, the effects of humidity forcing on forecast accuracy will need to be continually monitored.

#### **Supporting information**

S1 Data. 2003–2015 historical GFT estimates of municipal ILI, as these data were released in real time.

(CSV)

**S1 Text. Description of synthetic testing procedure and findings.** (DOCX)

**S1 Table.** Mean Friedman ranks of forecast error for predictions of synthetic truth targets of peak intensity, peak week and incidence during the first 2 weeks (RMSE2) and 4 weeks (RMSE4) of forecast. For pairwise tests of significance see S2 Table. Best performing model forms are in bold. Note, two forms may be best if not statistically different. (DOCX)

S2 Table. Pairwise p-values derived from Nemenyi tests of the synthetic forecast ranks shown in S1 Table. Asterisks designate differences significant at p<0.01 (\*\*) and p<0.001 (\*\*\*).

(DOCX)

**S1 Fig. Example forecasts from Atlanta, GA during 2005–2006.** Shown are the average trajectories across all 16 model-filter combinations for each of the 4 AH forcings. Successive weekly forecasts are shown for Weeks 45–60. Climatological AH forecasts are shown in red; combination AH forecasts in green; observed AH forecasts in blue; and no AH forecasts in purple. ILI+ observations are denoted by the black triangles and the vertical dashed line delimits the forecast initiation week and the observations used to optimize models. (TIFF)

**S2 Fig. Example forecasts from New York city during 2007–2008.** Shown are the average trajectories across all 16 model-filter combinations for each of the 4 AH forcings. Successive weekly forecasts are shown for Weeks 45–64. Climatological AH forecasts are shown in red; combination AH forecasts in green; observed AH forecasts in blue; and no AH forecasts in purple. ILI+ observations are denoted by the black triangles and the vertical dashed line delimits the forecast initiation week and the observations used to optimize models. (TIFF)

**S3 Fig. Example forecasts from Tampa Bay, FL during 2005–2006.** Shown are the average trajectories across all 16 model-filter combinations for each of the 4 AH forcings. Successive weekly forecasts are shown for Weeks 45–64. Climatological AH forecasts are shown in red; combination AH forecasts in green; observed AH forecasts in blue; and no AH forecasts in purple. ILI+ observations are denoted by the black triangles and the vertical dashed line delimits the forecast initiation week and the observations used to optimize models. (TIFF)

**S4 Fig. Percentage of forecasts accurate for predictions of peak intensity using various error margins to define accuracy.** Shown are the forecast accuracies as a function of week relative to predicted peak for models with climatological AH forcing (green), observed AH forcing (red), a combination of observed AH during optimization and climatological AH during forecast (blue), and no AH forcing (grey). The top left sub-panel shows accuracy per the error margins specified in the main manuscript. (TIFF)

**S5 Fig. Percentage of forecasts accurate for predictions of peak week using various error margins to define accuracy.** Shown are the forecast accuracies as a function of week relative to predicted peak for models with climatological AH forcing (green), observed AH forcing (red), a combination of observed AH during optimization and climatological AH during forecast (blue), and no AH forcing (grey). The top left sub-panel shows accuracy per the error margins specified in the main manuscript. (TIFF)

#### **Author Contributions**

Conceptualization: Jeffrey Shaman, Sasikiran Kandula.

Formal analysis: Jeffrey Shaman, Sasikiran Kandula.

Funding acquisition: Jeffrey Shaman.

Investigation: Jeffrey Shaman, Sasikiran Kandula, Alicia Karspeck.

Methodology: Jeffrey Shaman, Sasikiran Kandula, Wan Yang.

Project administration: Jeffrey Shaman.

Supervision: Jeffrey Shaman.

Visualization: Sasikiran Kandula.

Writing - original draft: Jeffrey Shaman.

Writing - review & editing: Jeffrey Shaman, Sasikiran Kandula, Wan Yang, Alicia Karspeck.

#### References

- 1. Loosli CG, Lemon HM, Robertson OH, Appel E. Experimental air-borne influenza infection. I. Influence of humidity on survival of virus in air. Proc Soc. Exper. Biol. Med. 1943; 53: 205–206.
- Shechmeister IL. Studies on the experimental epidemiology of respiratory infections. J. Infect. Dis. 1950; 87: 128–132. PMID: <u>14774528</u>
- 3. Hemmes JH, Winkler KC, Kool SM. Virus survival as a seasonal factor in influenza and poliomyelitis. Nature 1960; 188: 430–431. PMID: <u>13713229</u>
- 4. Harper GJ. Airborne micro-organisms: survival tests with four viruses. J. Hygiene 1961; 59: 479–486.
- Hemmes JH, Winkler KC, Kool SM. Virus survival as a seasonal factor in influenza and poliomyelitis. Antonie Van Leeuwenhoek 1962; 28: 221–233. PMID: 13953681
- 6. Hood AM. Infectivity of influenza virus aerosols. J. Hygiene 1963; 61: 331–335.
- 7. Mitchell CA, Guerin LF, Robillard J. Decay of influenza A viruses of human and avian origin. Canad. J. Comparative Medicine 1968; 32: 544–546.
- 8. Mitchell CA, Guerin LF. Influenza A of human, swine, equine and avian origin: comparison of survival in aerosol form. Canad. J. Comparative Medicine 1972; 36: 9–11.
- 9. Schaffer FL, Soergel ME, Straube DC. Survival of airborne influenza virus: effects of propagating host, relative humidity, and composition of spray fluids. Arch. Virology 1976; 51: 263–273.
- Yang W, Elankumaran S, Marr LC. Relationship between humidity and influenza A viability in droplets and implications for influenza's seasonality. PLoS ONE 2012; 7: e46789, https://doi.org/10.1371/ journal.pone.0046789 PMID: 23056454
- Lowen AC, Mubareka S, Steel J, Palese P. Influenza virus transmission is dependent on relative humidity and temperature. PLoS Pathog. 2007; 3: e151, https://doi.org/10.1371/journal.ppat.0030151 PMID: 17953482
- Shaman J, Kohn MA. Absolute Humidity Modulates Influenza Survival, Transmission and Seasonality. Proc. Nat. Acad. Sci. USA 2009; 106: 3243–3248, <u>https://doi.org/10.1073/pnas.0806852106</u> PMID: 19204283
- Koep TH, Enders FT, Pierret C, Ekker SC, Krageschmidt D, Kevin NL, et al. Predictors of Indoor Absolute Humidity and Estimated Effects on Influenza Virus Survival in Grade Schools. BMC Infectious Diseases 2013; 13: 71, https://doi.org/10.1186/1471-2334-13-71 PMID: 23383620
- Tamerius JD, Acosta LM, Jacobson JS, Goldstein IF, Quinn JW, Rundle AG, et al. Socioeconomic and Outdoor Meteorological Determinants of Indoor Temperature and Humidity in New York City Households. Weather, Climate and Society, 2013; 5: 168–179, https://doi.org/10.1175/WCAS-D-12-00030.1 PMID: 24077420
- Nguyen JL, Schwartz J and Dockery DW. The relationship between indoor and outdoor temperature, apparent temperature, relative humidity, and absolute humidity. Indoor Air 2014; 24: 103–112, <a href="https://doi.org/10.1111/ina.12052">https://doi.org/10.1111/ina.12052</a> PMID: 23710826

- Quinn A, Tamerius JD, Perzanowski M, Jacobson JS, Goldstein I, Acosta L and Shaman J. Predicting indoor heat exposure risk during extreme heat events. Sci. Total Environ. 2014; 490: 686–693, <u>https:// doi.org/10.1016/j.scitotenv.2014.05.039</u> PMID: 24893319
- Klepeis NE, Nelson WC, Ott WR, Robinson JP, Tsang AM, Switzer P, et al. The National Human Activity Pattern Survey (NHAPS): a resource for assessing exposure to environmental pollutants. J. Expo. Sci. Env. Epid. 2001; 11: 231–252, https://doi.org/10.1038/sj.jea.7500165 PMID: 11477521
- Shaman J, Pitzer VE, C. Viboud C, Grenfell BT, Lipsitch M. Absolute Humidity and the Seasonal Onset of Influenza in the Continental US. PLoS Biol. 2010; 8: e1000316, https://doi.org/10.1371/journal.pbio. 1000316 PMID: 20186267
- Barreca AI and Shimshack JP. Absolute humidity, temperature, and influenza mortality: 30 years of county-level evidence from the United States. Am J Epidem. 2012; 176: S114–S122, <u>https://doi.org/10. 1093/aje/kws259 PMID: 23035135</u>
- te Beest DE, van Boven M, Hooiveld M, van den Dool C and Wallinga J. Driving factors of influenza transmission in the Netherlands. Am J Epidem. 2013; 178: 1469–1477, https://doi.org/10.1093/aje/ kwt132 PMID: 24029683
- Tamerius JD, Shaman J, Alonso W, Bloom-Feshbach K, Uejio C, Comrie A and Viboud C. Environmental Predictors of Seasonal Influenza Epidemics across Temperate and Tropical Climates. PLoS Pathogens 2013; 9: e1003194, https://doi.org/10.1371/journal.ppat.1003194 PMID: 23505366
- 22. Shaman J, Goldstein E and Lipsitch M. Absolute humidity and pandemic versus epidemic influenza. Am J Epidem. 2011; 173: 127–135, https://doi.org/10.1093/aje/kwq347 PMID: 21081646
- Chowell G, Towers S, Viboud C, Fuentes R, Sotomayor V, Simonsen L, et al. The influence of climatic conditions on the transmission dynamics of the 2009 A/H1N1 influenza pandemic in Chile. BMC Infect. Dis, 2012; 12: 298, https://doi.org/10.1186/1471-2334-12-298 PMID: 23148597
- Yaari R, Katriel G, Huppert A, Axelsen JB and Stone L. Modelling seasonal influenza: the role of weather and punctuated antigenic drift. J. Royal Soc. Interface, 2013; 10: 20130298, <u>https://doi.org/10. 1098/rsif.2013.0298</u> PMID: 23676899
- Tamerius JD, Viboud C, Shaman J and Chowell G. Regional variability of specific humidity and school vacation can explain multiple spatially-focused waves of 2009 pandemic influenza in Mexico. PLOS Comp. Biol. 2015; 11: e1004337, https://doi.org/10.1371/journal.pcbi.1004337 PMID: 26291446
- Shaman J, Karspeck A. Forecasting seasonal outbreaks of influenza. Proc. Nat. Acad. Sci. USA 2012; 109: 20425–20430, https://doi.org/10.1073/pnas.1208772109 PMID: 23184969
- Shaman J, Karspeck A, Yang W, Tamerius J and Lipsitch M. Real-Time Influenza Forecasts during the 2012–2013 Season. Nature Communications 2013; 4: 2837, <u>https://doi.org/10.1038/ncomms3837</u> PMID: 24302074
- Yang W, Karspeck A and Shaman J. Comparison of filtering methods for the modeling and retrospective forecasting of influenza epidemics. PLOS Comp. Biol., 2014; 10: e1003583, <u>https://doi.org/10.1371/journal.pcbi.1003583</u> PMID: 24762780
- Shaman J and Kandula S. Improved Discrimination of Influenza Forecast Accuracy using Consecutive Predictions. PLOS Curr. Outbreaks, 2015; Oct 5. Edition 1, https://doi.org/10.1371/currents.outbreaks. 8a6a3df285af7ca973fab4b22e10911e PMID: 26512336
- Yang W, Cowling BJ, Lau EHY and Shaman J. Forecasting influenza epidemics in Hong Kong. PLOS Comp. Biol., 2015; 11: e1004383, https://doi.org/10.1371/journal.pcbi.1004383 PMID: 26226185
- 31. CPID-Columbia Prediction of Infectious Diseases. 2017 Available from: cpid.iri.columbia.edu
- Mesinger F, DiMego G, Kalnay E, Mitchell K, Shafran PC, Ebisuzaki W, et al. North American regional reanalysis. Bull. Amer. Meteor. Soc. 2006; 87: 343–360, https://doi.org/10.1175/BAMS-87-3-343
- Cosgrove BA, Lohmann D, Mitchell KE, Houser PR, Wood EF, Schaake JC, et al. Real-time and retrospective forcing in the North American Land Data Assimilation System (NLDAS) project. J. Geophys. Res. 2003; 108: D22, 8842, https://doi.org/10.1029/2002JD003118
- 34. Data Source: Google Flu Trends. 2015 Available from: http://www.google.org/flutrends.
- **35.** Data Source: Centers for Disease Control and Prevention FluView. 2016 Available from: <u>http://www.</u>cdc.gov/flu/weekly/
- Goldstein E, Viboud C, Charu V and Lipsitch M. Improving the Estimation of Influenza-related Mortality over a Seasonal Baseline. Epidemiology 2012; 23: 829–838, https://doi.org/10.1097/EDE. 0b013e31826c2dda PMID: 22992574
- Burgers G, van Leeuwen PJ and Evensen G. Analysis scheme in the ensemble Kalman filter. Mon. Wea. Rev. 1998; 126: 1719–1724, https://doi.org/10.1175/1520-0493
- Anderson JL. An ensemble adjustment Kalman filter for data assimilation. Mon. Wea. Rev. 2001; 129: 2884–2903, https://doi.org/10.1175/1520-0493

- Anderson JL. A Non-Gaussian Ensemble Filter Update for Data Assimilation. Mon. Wea. Rev. 2010; 138: 4186–4198, https://doi.org/10.1175/2010MWR3253.1
- Arulampalam MS, Maskell S, Gordon N and Clapp T. A Tutorial on Particle Filters for Online Nonlinear/ Non-Gaussian Bayesian Tracking. IEEE Trans. Signal Proc. 2002; 50:174–188, <u>https://doi.org/10.1109/78.978374</u>
- 41. Evensen G. Data Assimilation: The ensemble Kalman filter. 2nd ed. New York: Springer. 2009.
- 42. van Leeuwen PJ. Particle filtering in geophysical systems. Mon. Wea. Rev. 2009; 137: 4089–4114, https://doi.org/10.1175/2009MWR2835.1
- 43. Biggerstaff M, Jhung MA, Reed C, Fry AM, Balluz L, Finelli L. Influenza-like illness, the time to seek healthcare, and influenza antiviral receipt during the 2010–11 influenza season—United States. The Journal of infectious diseases. 2014; 210(4):535–544. https://doi.org/10.1093/infdis/jiu224 PMID: 24731959
- 44. Ginsberg J, Mohebbi MH, Patel RS, Brammer L, Smolinski MS, et al. Detecting influenza epidemics using search engine query data. Nature. 2009; 457:1012–1014. <u>https://doi.org/10.1038/nature07634</u> PMID: 19020500
- 45. Cook S, Conrad C, Fowlkes AL, Mohebbi MH. Assessing Google Flu Trends performance in the United States during the 2009 influenza virus A (H1N1) pandemic. PLoS One. 2011; 6:e23610. <u>https://doi.org/ 10.1371/journal.pone.0023610 PMID: 21886802</u>
- Olson DR, Konty KJ, Paladini M, Viboud C, Simonsen L. Reassessing Google Flu Trends Data for Detection of Seasonal and Pandemic Influenza: A Comparative Epidemiological Study at Three Geographic Scales. PLoS Comput Biol. 2013; 9(10): e1003256. <u>https://doi.org/10.1371/journal.pcbi.</u> 1003256 PMID: 24146603
- Lazer D, Kennedy R, King G, Vespignani A. The parable of Google Flu: traps in big data analysis. Science. 2014; 343(6176):1203–1205. https://doi.org/10.1126/science.1248506 PMID: 24626916
- Biggerstaff M, Alper D, Dredze M, Fox S, Fung ICH, Hickmann KS, et al. Results from the centers for disease control and prevention's predict the 2013–2014 Influenza Season Challenge. BMC Infectious Diseases. 2016; 16(1), 357.
- Tizzoni M, Bajardi P, Poletto C, Ramasco JJ, Balcan D, Gonçalves B, et al. Real-time numerical forecast of global epidemic spreading: case study of 2009 A/H1N1pdm. BMC medicine. 2012; 10(1):165.
- Brooks LC, Farrow DC, Hyun S, Tibshirani RJ, Rosenfeld R. Flexible modeling of epidemics with an empirical Bayes framework. PLoS computational biology. 2015; 11(8):e1004382. https://doi.org/10. 1371/journal.pcbi.1004382 PMID: 26317693
- Osthus D, Hickmann KS, Caragea PC, Higdon D, Del Valle SY. Forecasting seasonal influenza with a state-space SIR model. The Annals of Applied Statistics. 2017; 11(1):202–224. <u>https://doi.org/10.1214/ 16-AOAS1000 PMID: 28979611</u>
- Ray EL, Sakrejda K, Lauer SA, Johansson MA, Reich NG. Infectious disease prediction with kernel conditional density estimation. Statistics in Medicine. 2017;1–22. https://doi.org/10.1002/sim.7488